Author’s response to reviews

Title: Radiographic 2D:4D Index in Females: No Relation to Anthropometric, Behavioural, Nutritional, Health-Related, Occupational or Fertility Variables

Authors:

Tapio Vehmas (tapio.vehmas@ttl.fi)
Svetlana Solovieva (svetlana.solovieva@ttl.fi)
Paivi Leino-Arjas (paivi.leino-arjas@ttl.fi)

Version: 2 Date: 2 August 2006

Author’s response to reviews:

Journal of Negative Results in Biomedicine

Dear Editor

Thank you once again for the encouraging comments concerning our manuscript MS: 1674784144105020 "Radiographic 2D:4D Index in Females: No Relation to Anthropometric, Behavioural, Nutritional, Health-Related, Occupational or Fertility Variables". We have modified the text according to the given advice:
* We now indicate that this is a secondary study (Background: last paragraph).
* Figure 3 has been omitted and text added: "The symmetrically distributed 2D:4D index ranged..." (Results, 1st line).
* Tables 1-2 defining the clusters have been removed. New tables 1-2 have been added instead and they show the relation between the 2D:4D index and the relevant individual variables before clustering. Relevant explanations have been added in the text.
We hope that the text is now satisfactory.

Helsinki 2nd Aug 2006

Sincerely

Tapio Vehmas, MDPhD, Chief radiologist

Finnish Institute of Occupational Health
Topeliuksenkatu 41 a A, FIN-00250 Helsinki, Finland
tel. +358-30-4742481, +358-400-640496 (mobile)
FAX +358-30-4742108, +358-9-2412414